

OXFORD

Clinical science

Dual biologic therapy for the treatment of rheumatic diseases and asthma: a case series

Mariam Malik¹, Bryony Jones¹, Emma Williams², Ramesh Kurukulaaratchy³*, Chris Holroyd¹, Alice Mason¹

¹Rheumatology Department, University Hospital Southampton, Southampton, UK

²Rheumatology Department, Royal Hampshire County Hospital, Winchester, UK

³Department of Respiratory Medicine, University Hospital Southampton, Southampton, UK

*Correspondence to: Ramesh Kurukulaaratchy, Department of Respiratory Medicine, Southampton General Hospital, Mailpoint 810, F-Level, South Academic Block, Tremona Road, Southampton S016 6YD, UK. E-mail: r.j.kurukulaaratchy@soton.ac.uk

Abstract

Objective: Combination biological therapies are being considered increasingly for patients with multiple co-morbidities requiring biologics. There are limited data available on this approach, and concerns remain about the possible risk of adverse events, particularly infection.

Methods: We present three patients on dual biologics for rheumatic disease and asthma. The biologic combinations used were etanercept and mepolizumab, infliximab and omalizumab, and etanercept and omalizumab. The time on combination biologic therapies ranged from 24 to 36 months. Patients were monitored for any serious adverse events.

Results: All three patients were able to tolerate combined biologic therapies, with no serious adverse events. All three patients gained improvement in their rheumatic and asthma disease control, with reduction in disease activity scores and reduction in steroid usage.

Conclusion: The decision to start dual biologic therapy should be considered carefully, on a case-by-case basis. The number of patients who are on combination biological therapy is small, and data are sparse. Real-world data are needed to examine the long-term benefits and risks of different forms of combination biologic therapies.

Lay Summary

What does this mean for patients?

Biological therapies are specific therapies that target particular parts of the immune system. These medications are used in the treatment of autoimmune conditions, such as rheumatoid arthritis (RA). In RA, they work by stopping particular chemicals in the blood from activating the immune system to attack joints. They are very effective medications to treat inflammatory autoimmune conditions. In some cases, they can be associated with side effects, particularly increased risk of infections. We have written a case series about three people who have inflammatory autoimmune conditions and are receiving two biologics to treat these. All three people have inflammatory arthritis and asthma and are receiving two biologics to treat these. All three people have inflammatory arthritis and asthma and are receiving two biologics to treat these. All three people have inflammatory arthritis and asthma and are receiving two biologics to treat these. All three people have inflammatory arthritis and asthma and are receiving two biologics to treat these. All three people have inflammatory arthritis and asthma and are receiving two biologics to treat these. All three people have inflammatory arthritis and asthma control, with reduction in disease activity and the use of steroids. Although the number of people receiving more than one biological treatment is small, this study suggests that dual therapy could be used in certain situations. The decision to start dual biologic therapy should be considered carefully, on a case-by-case basis, by the specialists.

Keywords: rheumatic disease, asthma, immunosuppressants, biological therapies, infections and arthritis, quality of life, etanercept, mepolizumab, infliximab, omalizumab

Key messages

- Combination biological therapies are being considered increasingly for patients who present with multiple co-morbidities.
- Data presented here suggest that combination biologics can be used safely, without serious adverse events.
- The decision to start dual biologic therapy should be considered carefully, on a case-by-case basis.

Introduction

Combination biological therapies are being considered increasingly for the treatment of patients who present with multiple co-morbidities requiring biological treatment. There are limited data available regarding this, and concerns remain about the risk of adverse events, particularly infections, in addition to potential drug interactions in patients receiving more than one biological therapy [1]. We present a case series of three patients who have benefited from dual biologic therapy for rheumatic disease and asthma.

Received: 7 July 2022. Accepted: 17 January 2023

[©] The Author(s) 2023. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Literature on dual biologic therapy is limited, owing to the infrequency of this clinical approach. There are significant clinical reservations in taking this approach owing to the potential risk of serious infection from immunosuppression [1]. In addition, the wide variety of biological therapies and their potential combinations makes consistent comparison of outcomes difficult, and structured clinical trials are infrequent.

Development of newer agents targeting different pathways has made it possible to use combination therapy in selected patients. Simultaneous use of more than one biological agent has been used in two main scenarios: patients with severe inflammatory disease unresponsive to a single biological agent, or patients with co-existing severe medical conditions, such as Inflammatory Bowel Disease [2–4], asthma [5, 6] and psoriasis [7]. There is limited real world evidence from case-based literature on combining biological therapies across different clinical specialties.

Methods

We present three cases illustrating the outcome of combination therapy for severe asthma and inflammatory arthritis. Formal ethics approval was not required for this work because only anonymized patient information was included and no change to patients' standard care was required in order to be included in the case series. All patients gave written informed consent to the use of their data. A summary of the clinical presentations and therapies can be found in Table 1.

Etanercept and infliximab are anti-TNF- α agents licensed for the treatment of moderate or severe RA [8]. Omalizumab is an anti-IgE biologic used for the treatment of atopic asthma [9]. Mepolizumab is an anti-IL-5 agent used for the treatment of severe eosinophilic asthma.

We have defined serious adverse events in accordance with the US Food and Drug Administration definition [10]. A serious adverse event is an undesirable experience associated with the use of a medical product, which is life threatening, can result in death, hospitalization, disability, permanent damage or congenital anomaly. This also includes other serious events, such as allergic bronchospasm, serious blood dyscrasias or seizures. We have also included transaminitis and infections requiring antibiotics. Information was gathered by reviewing medical notes at the tertiary centre, the local hospital, from general practitioner records and by contacting patients.

Case one

Case one is a 56-year-old woman with severe eosinophilic asthma diagnosed in the late 1990s and seronegative RA diagnosed in 2019. Her asthma was previously treated with

Table 1. Summary of key points of each case

bronchodilator and steroid inhalers, montelukast and daily maintenance prednisolone. Mepolizumab was commenced in 2018, and since then she has reported no exacerbations of her asthma. Mid-2022, she was reviewed by the asthma service and was tolerating mepolizumab well, was no longer on montelukast or the steroid inhaler, and had not had to use her salbutamol inhaler for \sim 7 months.

For RA, she was initially treated with sulphasalazine (SSZ), but this was not tolerated and was stopped. She commenced methotrexate (MTX) 20 mg s.c. once weekly along with prednisolone in early 2020, but after 6 months remained symptomatic, with evidence of active synovitis on musculoskeletal ultrasound (US), despite being on prednisolone 25 mg daily. At this time, her disease activity score (DAS28) was 6.25.

Etanercept was then commenced alongside MTX and prednisolone. Twenty-four months after commencing etanercept, she continues etanercept and has reduced her prednisolone to 11 mg daily. Her DAS is 2.27. It has been difficult to reduce her prednisolone further thus far owing to ongoing joint pain, although there is no synovitis on US. She is being encouraged to reduce the corticosteroids (CS) dose gradually. She has been on dual therapy for 24 months and has not had any recorded serious adverse events.

Case two

Case two is a 35-year-old man, known to have asthma since childhood, requiring multiple hospital admissions and an intensive care unit admission in 2002. He has previously been on a combination of inhalers, aminophylline and montelukast, but still required frequent courses of oral CSs. He was diagnosed with Crohn's disease in 2006 and was subsequently diagnosed with Crohn's-associated arthritis, for which he has been on azathioprine (AZA) and infliximab since 2008. In April 2019 he was started on omalizumab for severe asthma that was difficult to control. On follow-up, he reported significant improvement in asthma symptoms, with no further need for oral CSs, and he no longer requires nebulized therapy. He continues infliximab at the same dose and omalizumab and has achieved good control of his Crohn's disease and associated arthritis. He has not experienced any serious adverse effects after 36 months of combination therapy.

Case three

Case three is a 62-year-old woman who was diagnosed with atopic eosinophilic adult-onset asthma in the late 1980s and seropositive RA in 2012. Prior therapy for asthma included a combination of bronchodilator and steroid inhalers. She was referred to the severe asthma service in 2018, and

Age (years)	Diagnosis	Medications (biologics)	Serious adverse events	Rheumatic disease control
56	Seronegative RA Asthma	Etanercept, 2020 Mepolizumab, 2018	Nil	Continues prednisolone 11 mg with no evidence of active synovitis on US, although it has been difficult to reduce prednisolone further
35	Crohn's-associated arthritis Crohn's disease Asthma	Infliximab, 2008 Omalizumab, 2019	Nil	Remission
62	Seropositive RA Asthma	Omalizumab, 2019 Etanercept, 2018	Nil	Remission

omalizumab was commenced in June 2019. Since starting this, her asthma has been well controlled, supported by clinical examination and ACQ-6 score (asthma control questionnaire). Initially, her RA was treated with MTX and hydroxychloroquine (HCQ). However, despite this combination she had evidence of ongoing joint inflammation, and in 2018 she was started on etanercept. She has continued both biologics for 36 months with good control of asthma and RA, without any serious adverse events, and has not required any steroid therapy. This was confirmed by reviewing the omalizumab record sheet.

Results

In these three cases, dual biologic therapy has been used to good effect, with no infection or other serious adverse events. This has also enabled patients to reduce oral CS dependence significantly, with the added benefit of minimizing CS-related adverse effects.

Discussion

The decision to start a patient on dual biologic therapy should be considered carefully, on a case-by-case basis, only when conventional treatment options have failed and in the context of life-threatening or significantly disabling disease. The main concerns regarding the use of dual biologic therapy centre around safety, especially a theoretical risk of serious infections [1]: no infections were seen in this small case series. This potential risk must be balanced carefully against the potential long-term harm from ongoing severe disease and the welldocumented risks associated with other adjuvant therapies, specifically CSs, which would be the most likely alternative treatment option in this group of difficult-to-treat patients. The number of patients requiring combination biological therapies is a small but important group. Data are sparse, because they are generally excluded from clinical trials. Thus, real-world data are important and needed to examine the long-term benefits and risks of different forms of combination biologic therapies across different disease areas.

Data availability

The data underlying this article cannot be shared publicly owing to ethical/privacy reasons.

Funding

No specific funding was received from any bodies in the public, commercial or not-for-profit sectors to carry out the work described in this article.

Disclosure statement: E.W. has received speaker fees/honoraria from AbbVie, Internis, Novartis, Nordic Pharma, Roche, Thornton & Ross and UCB and contributed to advisory boards for Lilly. C.H. has received honoraria or sat on advisory boards for Abbvie, BMS, Celltrion, Chugai, Galapagos/Gilead, Janssen, Lilly, Napp, Novartis, Pfizer and UCB. The remaining authors have declared no conflicts of interest.

References

- Boleto G, Kanagaratnam L, Dramé M, Salmon JH. Safety of combination therapy with two bDMARDs in patients with rheumatoid arthritis: a systematic review and meta-analysis. Semin Arthritis Rheum 2019;49:35–42.
- 2. Ahmed W, Galati J, Kumar A *et al.* Dual biologic or small molecule therapy for treatment of inflammatory bowel disease: a systematic review and meta-analysis. Clin Gastroenterol Hepatol 2022;20: e361–9.
- 3. Gold SL, Steinlauf AF. Efficacy and safety of dual biologic therapy in patients with inflammatory bowel disease: a review of the literature. Gastroenterol Hepatol (NY) 2021;17:406–14.
- Ribaldone DG, Pellicano R, Vernero M *et al.* Dual biological therapy with anti-TNF, vedolizumab or ustekinumab in inflammatory bowel disease: a systematic review with pool analysis. Scand J Gastroenterol 2019;54:407–13.
- Eggert L, Chinthrajah RS. Switching and combining biologics in severe asthma: Experience from a large academic teaching center. https://doi.org/10.1164/ajrccm-conference.2019.199.1_Meeting Abstracts.A1309.
- Thomes R, Darveaux J. Combination biologic therapy in severe asthma: a case series. Ann Allergy Asthma Immunol 2018;121:S91.
- Mahar PD, Zubrinich CM, Manuelpillai N, Foley P. Combination treatment with monoclonal antibodies: secukinumab, benralizumab and dupilumab for the combined management of psoriasis and severe asthma. Australas J Dermatol 2021;62:506–8.
- Holroyd CR, Seth R, Bukhari M *et al.* The British Society for Rheumatology biologic DMARD safety guidelines in inflammatory arthritis. Rheumatology 2019;58:e3–42.
- National Institutes of Health. Global Initiatives for Asthma. Global strategy for asthma management and prevention. 2018. www. ginasthma.org (13 April 2022, date last accessed).
- US Food and Drug Administration. FDA. What is a serious adverse event? https://www.fda.gov/safety/reporting-serious-problems-fda/ what-serious-adverse-event (13 April 2022, date last accessed).



A 2nd generation, JAK1 preferential inhibitor for moderate to severe RA¹⁻⁶

While 1st generation JAK inhibitors are relatively non-selective,²⁻⁶ JYSELECA has over 5x greater potency for JAK1 over JAK2/3 and TYK21*

Balancing sustained efficacy⁷⁻¹¹ with acceptable tolerability^{1,12}



()

Indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti-rheumatic drugs.¹ May be used as monotherapy or in combination with methotrexate.¹

*From biochemical assays, the clinical relevance of which is uncertain. JAK, Janus kinase; RA, rheumatoid arthritis; TYK, tyrosine kinase.

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information.

۲

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information. **JYSELECN** figotinib 100 mg or 200 mg film-coated tablets. **Indication:** Jyseleca is indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti rheumatic drugs (DMARDs). Jyseleca may be used as monotherapy or in combination with methotrexate (MTX). **Dosage:** <u>Adults</u>; 200 mg once daily. Taken orally with/without food. It is recommended that tablets are swallowed whole. <u>Laboratory Monitoring:</u> Refer to the SmPC for information regarding laboratory monitoring and dose initiation or interruption. <u>Elderly</u>, 4 starting dose of 100 mg of filgotinib once daily is recommended for patients aged 75 years and older as clinical experience is limited. <u>Renal impairment</u>: No dose adjustment required in patients with estimated creatinine clearance (CrCl) ≥ 60 mL/min. A dose of 100 mg of filgotinib once daily is recommended for patients with estimated and by is recommended the patients with estimated creatinine clearance (CrCl) ≥ 60 mL/min. A dose of 100 mg of seadjustment required. Severe hepatic impairment: not dose adjustment required. Severe hepatic impairment: not dose adjustment required. Severe hepatic impairment: not dose adjustment required. Severe hepatic impairment: not geomended. <u>Children (< 18years</u>): Safety and efficacy not yet established. **Contraindications**: Hypersensitivity to the active substance or to any of the excipients. Active tuberculosis (TB) or active serious infections. Pregnancy. **Warnings/Precautions**: information use, with immunosuppressions (and be excluded. <u>Infections</u>; infections, including serious infections edidisis; and cryptococcosis have been reported. Risk beneft should be assessed prior to initiating in patients with risk factors for infections (see SmPC). Patients should be closely monitored for the development of theatenes. have been reported, Kisk benefit should be assessed phore of hitating in patients with risk factors for infections (see SmPC). Yatients should be closely monitored for the development of igns and symptoms of infections during and after fligotinib reatment. Treatment should be interrupted if the patient

is not responding to antimicrobial therapy, until infection is controlled. There is a higher incidence of serious infections in the elderly aged 75 years and older, caution should be used when treating this population. <u>Tuberculosis</u>, Patients should be screened for TB before initiating filgotinib, and filgotinib should not be administered to patients with active TE. <u>Viral</u> <u>reactivation</u>: Cases of herpes virus reactivation (e.g., herpes zoster), were reported in clinical studies (see SmPC). If a patient develops herpes zoster, filgotinib treatment should be temporarily interrupted until the episode resolves. Screening for viral hepatitis and monitoring for reactivation should be performed. <u>Malignancy</u>: Immunomodulatory medicinal products may increase the risk of malignancies. Malignancies were observed in clinical studies (see SmPC). <u>Ferlility</u>. In animal studies, decreased ferlility, impaired spermatogenesis, and histopathological effects on male reproductive organs were observed (see SmPC). The potential effect of filgotinib on sperm production and male fertility in humans is currently unknown. <u>Haematological abnormalities</u>; Do not start therapy, or temporarily stop, if Absolute Neutrophil Count (ANC) 1< 10° (cells/L, ALC - OS + 10° cells/L or chaemoglobin «B g/dL. Temporarily stop therapy if these values are observed during routine patient management. <u>Vaccinations</u>; Use of live vaccines during, or immediately prior to, filgotinib treatment is not recommended. <u>Lipids</u>: Treatment with filgotinib was associated with dose dependent increases in lipid parameters, including total cholesterol, and high-density lipoprotein (LDL) levels, while low density lipoprotein (LDL) levels were slightly increased (see SmPC). <u>Cardiovascular</u> *tisk*; Rheumatoid arthritis patients have an increased insk for cardiovascular disorders. Patients should have risk factors (e.g., hypertension, hyperlipidaemia) managed as part of usual standard of care. <u>Venous thrombobembolism</u>: Events of deep venous thrombosis (OVT) and pulmona of DVT/PE, or patients undergoing surgery, and prolonged

Learn more at strengthofbalance.co.uk

immobilisation. Lactose content: Contains lactose; patients with rare hereditary problems of galactose intolerance, total lactase deficiency or glucose-galactose malabsorption should not take filgotinib. **Pregnancy/Lactation**: Filgotinib is contraindicated in pregnancy. Filgotinib should not be used during breast-feeding. Women of childbearing potential must use effective contraception during and for at least 1 week after cessation of treatment. **Driving/Using machinery**: No or negligible influence, however dizziness has been reported. **Side effects:** See SmPC for full information. <u>Common (21/100)</u>: herpes zoster, pneumonia, neutropenia, hypercholesterolasemia infection and dizziness. <u>Uncommon (s1/1000 to 1/100)</u>; herpes zoster, pneumonia, neutropenia, hypercholesterolaemia and blood creatine phosphokinase increase. Serious side effects: See SmPC for full information **Legal category**: POM **Pack**: 30 film-coated tablets/bottle **Price**: UK Basic NHS cost: £863.10 Marketing authorisation number(5): <u>Great Britain</u> Jyseleca 100mg film-coated tablets PLGB 42147/0002 hypeleca 200mg film-coated tablets PLGB 42147/0002 hypeleca 200mg film-coated tablets PLGB 42147/0002 hypeleca 100mg film-coated tablets EU/1/20/1480/001 EU/1/20/1480/003 EU/1/20/1480/004 **Further information**: Galapagos UK, Belmont House, 148 Belmont Road, Uxbridge (DB8 105, United Kingdom 00800 7387 1345 **medicalinfo@glgg**. <u>com</u> Jyseleca[®] is a trademark. **Date of Preparation**: January 2022 UK-RA-FIL-202201-00019 **W** Additional monitoring required Additional monitoring required

Adverse events should be reported. For Great Britain and Northern Ireland, reporting form and information can be found at <u>yellowcard.mhra.gov.</u> and information can be found at <u>yellowcard.mnra.gov.u</u> or via the Yellow Card app (download from the Apple Ap Store or Google Play Store). Adverse events should also be reported to Galapagos via email to DrugSafety.UK.Ireland@glpg.com or 00800 7878 1345

References: 1. JYSELECA SPC. Available at: www.medicines.org.uk. Last accessed: June 2022. 2. Angelini J, et al. Biomolecules 2020;10(7):E1002. 3. Banerjee S, et al. Drugs 2017;77:521–546. 4. O'Shea JJ, et al. Nat Rev Rheumatol 2013;9(3):173–182. 5. Traves PG, et al. Ann Rheum Dis 2021;0(1-11. 6. McInnes IB, et al. Arthr Res Ther 2019;21:183. 7. Combe B, et al. Ann Rheum Dis 2021;doi:10.1136/ annrheumdis-2020-219214. 8. Genovese MC, et al. JAMA 2019;322 (4):315–325. 9. Westhowers R, et al. Ann Rheum Dis 2021;doi:10.1136/annrheumdis-2020-219213. 10. Combe B, et al. Arthritis Rheumatol 2021;73(suppl 10). https://acrabstract/clinical-ou-week-48-of-fig0tinib-treatment-in-an-ongoing-long-term-extension-trial-of-a-plates-twith-inadequate-response-to-mtx-initially-treated-with-filgotinib-ra-adalimumab-during-th/. Last accessed: June 2022. 11. Buch MH, et al. Arthritis Rheumatol 2021;73 (suppl 10). https://acrabstracts.org/abstract/clinical-outcomes-up-to-week-48-of-nogoing-filgotinib-tra-abstracts.org/abstract/Jactastracts.org/abstract/sorg/abstract/linical-outcomes-up-to-week-48-of-Ingoing-filgotinib-tra-abstracts.org/abstract/linical-outcomes-up-to-week-48-of-nogoing-filgotinib-tra-abstracts.org/abstract/linical-outcomes-up-to-week-48-of-ongoing-filgotinib-tra-abstracts.org/abstract/lineg/abstract/linegrated-safety-analysis-update-for-filgotinib-in-patients-with-moderately-to-severely-active-rheumatoid-arthritis-receiving-treatment-over-a-median-of-2-2-years/. Last accessed: June 2022.



June 2022 GB-RA-JY-202205-00033

JYSELECA, GALAPAGOS and the JYSELECA and GALAPAGOS logos are registered trademarks of Galapagos NV. © 2022 Galapagos NV. All rights reserved.

۲